DETERMINE Working Document # 4

ECONOMIC ARGUMENTS

for addressing social determinants of health inequalities

DETERMINE – an EU Consortium for Action on the Socio economic Determinants of Health



Summary

DETERMINE is an EU Consortium for Action on the Socioeconomic Determinants of Health (SDH). The overall objective of DETERMINE is to achieve greater awareness and capacity amongst decision makers in all policy sectors to take health and health equity into consideration and to strengthen collaboration between health and other sectors.

DETERMINE consists of seven complementary but discrete work packages. This working document summarises the second year activities of work package 5. It explores if and how economic arguments are being used to support action at EU level and in 12 countries and regions across Europe.

There are strong economic arguments for investing in health at population level and the benefits of reducing health inequalities are beginning to be recognised in economic terms. Furthermore, the recent Communication from the European Commission *Solidarity in Health: Reducing health inequalities in the EU* is an indication of a focus on health inequalities at EU level.

The evidence presented here paves the way for further exploration at EU and member state level to analyse in economic terms the health impacts of policies that address the social determinants of health. Such work can help to provide further evidence to support the argument that targeted investment to address health inequalities by action on the social determinants of health is more cost effective than paying for the consequences of health inequalities.

Acknowledgments

This report is a product of the project DETERMINE: an EU consortium for Action on the Socioeconomic Determinants of Health. It was written by Teresa Lavin and Owen Metcalfe from the Institute of Public Health in Ireland. The authors would like to thank the following for their contributions:

- Ingrid Stegeman, EuroHealthNet
- Veerle Stevens, VIGeZ Flemish Institute for Health Promotion, BELGIUM
- Eve-Mai Rao and Tiia Pertel, National Institute for Health Development, ESTONIA
- Marita Sihto, THL National Institute for Health and Welfare, FINLAND
- Jorlaug Heimisdottir, Public Health Institute of Iceland, ICELAND
- Joop ten Dam and Janine Vervoordeldonk, NIGZ Netherlands Institute for Health Promotion and Disease Prevention, NETHERLANDS
- Claire Higgins, Institute of Public Health in Ireland, NORTHERN IRELAND¹
- Elisabeth Fosse, HEMIL Research Centre for Health Promotion, University of Bergen, NORWAY
- Justyna Car, National Institute of Hygiene, POLAND
- Sheila Beck, NHS Health Scotland, SCOTLAND
- Igor Krampac and Olivera Stanojevic, Regional Public Health Institute Maribor, SLOVENIA
- Sara Darias Curvo, University de La Laguna, Tenerife, SPAIN
- Malcolm Ward, Wales Centre for Health, WALES

The authors would also like to thank Marc Suhrcke, Peter Smith and Silvia Evers for helpful comments received in developing the template and on the final paper.

DETERMINE is coordinated by EuroHealthNet, in collaboration with the National Institute of Public Health, Czech Republic. This project has received co-funding from the European Commission, in the framework of the Public Health Programme.

DETERMINE Working Document

December 2009

www.health-inequalities.eu

¹ The Institute of Public Health in Ireland works across the two jurisdictions of Northern Ireland and the Republic of Ireland

Contents

Summary	2	
Acknowledgments	3	
Introduction	5	
I. Instrument design and data collection	6	
2. Making the case: Establishing economic arguments	8	
3. Opportunities and challenges to using economic arguments		
4. Examples of economic evaluations that consider health outcomes	17	
5. Conclusion and recommendations	21	
References	24	
Appendix I: Glossary of frequently used terms	29	
Appendix 2:Task timeline & questionnaire	30	

Introduction

DETERMINE is a three year project (2007-2010) supported by the European Commission and coordinated by EuroHealthNet. It brings together a high level Consortium with representation from 26 European countries with an overall objective of supporting and enabling policy makers and practitioners in all policy sectors to place a higher priority on health and health inequalities when developing policy. A core aim of DETERMINE is to strengthen the knowledge base on policies and actions addressing social determinants of health inequalities. DETERMINE consists of seven discrete but complementary work packages. The specific contributions of work package 5 are:

- Year 1: Identification and exploration of member state and EU level approaches to addressing social determinants of health inequalities.
- Year 2: Identification and exploration of economic arguments to support this work.

This working document summarises the second year activities of work package 5. For information on the first year activities of work package 5 and the DETERMINE project please visit <u>www.health-inequalities.eu</u>

Chapter one describes how the task was developed and the data collected. Chapter two provides the findings of a literature review of economic arguments for addressing social determinants of health inequalities. Reports from DETERMINE partners are presented in chapters three and four: Chapter three presents the opportunities and challenges to using economic arguments to address health inequalities while chapter four presents examples of economic evaluations conducted at member state and EU level that have considered health outcomes. Chapter five offers concluding remarks and recommendations. A glossary of frequently used terms is provided in Appendix 1 while Appendix 2 gives additional information on the task design.

I. Instrument design and data collection

Introduction

This chapter provides an overview of how the framework for identifying and exploring economic arguments for addressing social determinants of health inequalities was developed. It also describes how the data was collected and provides general comments on responses received.

Aim and objectives

The aim of this task was to better understand if and how economic arguments are being used to address social determinants of health inequalities within selected member states and by the institutions of the European Union. It is intended to use findings from this task to achieve more widespread adoption of effective and proven approaches to tackling health inequalities.

The objectives of the task were:

- To explore opportunities and challenges to using economic arguments to address social determinants of health inequalities.
- To identify examples of relevant economic evaluations that consider outcomes in terms of health and health inequalities.

Contributors

Members of the DETERMINE Consortium with a specific interest in work package 5 were asked to assess the situation within their own country or region. Partners representing 12 countries and regions across Europe undertook assessments and EuroHealthNet reviewed the situation at EU level. General contributions were received from additional members of the DETERMINE Consortium. The task was coordinated by the Institute of Public Health in Ireland.

Development of the guidance document & questionnaire

Initial scoping of the task took place at the second DETERMINE Consortium meeting in May 2008. Following this meeting, a review of relevant literature was undertaken and teleconferences were held with three key informants to clarify what is already known about economic arguments to address social determinants of health inequalities and how the DETERMINE project could complement and progress this work. This helped inform development of a draft questionnaire which was discussed and reviewed with partners at a workshop held in August 2008. At the workshop, one issue identified was the extent to which different policy areas should be included. It was agreed that the focus would be on policies driven by sectors other than health. However public health policies related to, for example, smoking, obesity and physical activity would be included but other healthcare polices, including treatment and services, would be excluded. The first part of the questionnaire sought to identify examples of economic evaluations and then explore how they came about, their challenges and their successes. The second part of the questionnaire was designed to explore the economic arguments being used, who the main drivers were and who such arguments were intended for. If there was limited debate around such issues, partners were asked to comment on why they thought this may be so and to include their suggestions for future action in this area.

Data collection

Data collection took place between September 2008 and January 2009. Partners were responsible for collecting data within their own country or region using the questionnaire template. EuroHealthNet, in collaboration with the work package leader, coordinated the collection of data at the EU level. It was agreed that given time and resource limitations partners were not expected to undertake a comprehensive review of all policies in all sectors.

A total of thirteen completed questionnaires were received. Another partner contributed general comments but was unable to complete the questionnaire. Most partners indicated that they used a combination of their own knowledge and the expertise of colleagues and others to complete the questionnaire. A wide variation was noted in the depth of responses with several partners reporting difficulty in identifying any relevant examples of economic evaluations. Overall, more information was provided exploring the opportunities and challenges to progressing this approach than actual examples of economic evaluations. It is outside the scope of this task to assess the extent to which examples reported and issues highlighted reflect overall activity in this area in the different regions and countries.

Findings reported in the following chapters must be interpreted giving due consideration to the limited data available on this area of work, the limited expertise devoted to addressing this issue in many countries as well as limited resources available for this part of the project. The authors view these findings and the report as a useful first step in addressing this complex and complicated area.

2. Making the case: Establishing economic arguments

Introduction

This chapter presents the findings of a literature review conducted in June 2008 together with further research in April 2009. The review was undertaken to help inform and shape further data collection by establishing the main economic arguments for addressing social determinants of health inequalities.

Health can be considered in economic terms as both a capital and a consumption good. In the case of health as a capital good, people in good health attract a higher value than those in poor health due to their greater ability to be economically productive. Health as a consumption good is concerned with the contribution that good health makes to an individual's wellbeing, happiness or satisfaction.

Targeted investment to address health inequalities by action on social determinants of health is more cost effective than paying later for the consequences of these inequalities. It follows then that addressing health inequalities is not only a matter of social justice but also contributes to economic growth.

The economic benefits of better health

There are strong economic arguments for investing in health at population level. In 2001, the WHO *Commission on Macroeconomics and Health* demonstrated that a healthier population can bring substantial economic benefits to countries with developing economies. The authors estimated that 50% of the growth differential between rich and poor countries was due to ill health and shorter life expectancy. Increasing life expectancy at birth by 10% through targeted investment to tackle the major causes of premature deaths could increase the economic growth rate by 0.35% per year. (1)

Benefits to the economy may not be limited to developing countries. In 2005, a review of evidence in high income countries concluded that "there is considerable and convincing evidence that significant economic benefits can be achieved by improving health not only in developing but also in developed countries." The report outlines four channels through which this may occur: higher productivity, higher labour supply, improved skills as a result of greater education and training and increased savings available for investment in physical and intellectual capital. (2) However more recent work suggests that the evidence is less conclusive about the links between health and the economy particularly at the macro level. (3)

These issues have been considered in the development of a health strategy for the EU which commenced with a reflection process in 2004. (4) The second of four principles in the strategy *Together for Health* "Health is the greatest wealth" states that health is a prerequisite for economic productivity and prosperity. It also emphasises that healthy life expectancy not merely life expectancy is the key factor for economic growth and points to the inclusion of Healthy Life Years as a Lisbon Structural Indicator to underline this distinction. (5) The role of healthcare systems in supporting growth and employment has been recognised in a report commissioned as part of the mid-term review of the Lisbon strategy. *Facing the challenge:The Lisbon strategy for growth and employment* notes the importance of healthcare systems "... in generating social cohesion, a productive workforce, employment and hence economic growth." (6)

The economic costs of ill health

Clear understanding of the cost of ill health is a prerequisite for assessing the economic returns of investing in health. While healthcare costs are substantial and increasing, these represent only one part of the overall picture.

Failure to include the welfare costs of ill health risks understating the true economic benefits derived from health interventions. Three perspectives have been identified to gain a more complete picture of the costs of ill health. The broad perspective encompasses social welfare costs; the limited perspective considers micro- and macroeconomic costs; while the very limited perspective focuses on healthcare costs. (7)

Cost of illness studies separate costs into three components: direct costs, i.e. those associated with treating illness; indirect costs, i.e. costs associated with loss of productivity due to morbidity or premature death and; intangible costs which include the psychological dimensions of illness. However in many cases only the first two are measured. A review of cost of illness studies found that the cost of chronic diseases ranged from 0.02% to 6.77% of a country's GDP. Cardiovascular disease in particular was found to account for between 1-3% of GDP in most developed countries. (8)

The cost of health inequalities

Health inequalities are understood to be the difference in health outcomes between different population groups, including socioeconomic groups. Such inequalities are estimated to reduce average life expectancy across the EU25² by 1.84 years which equates to approximately 11.4 million life years lost. At the same time, healthy life expectancy is reduced due to the existence of inequalities by an average of 5.14 years or approximately 33 million healthy life years lost. (9) While these figures represent the EU25 as a whole, the large variation in the magnitude of health inequalities within member states is well recognised, with some southern European countries having smaller inequalities and most countries in the East and Baltic regions having larger inequalities than the European average. (10)

These inequalities have significant economic implications for the EU and for member states. When health is valued as a capital good, inequalities related losses have been estimated to cost around \in 141 billion in 2004 or 1.4% of GDP. This rises sharply to \in 1,000 billion or 9.5% of GDP when health is valued as a consumption good. (9)

² These calculations were based on 2004 membership of the EU (25 member states)

Addressing health inequalities through action on social determinants of health

Just as the costs to society of ill health go beyond the cost of treatment, preventing ill health and reducing health inequalities is not the responsibility of the health care system alone. Health and health inequalities are largely determined by factors outside the reach of the healthcare sector, including low income, unemployment, poor environment, poor education and sub-standard housing. (11) Improving health and reducing health inequalities therefore requires action and investment across government to tackle these root causes. (2)

The *Tallinn Charter*, signed at the WHO European Ministerial Conference on Health Systems in June 2008, reiterates that "…investing in health is investing in human development, social well-being and wealth." It includes a commitment from member states to "…invest in health systems³ and foster investment across sectors that influence health, using evidence on the links between socioeconomic development and health." (12)

Following the report of the WHO Commission on Social Determinants of Health *Closing the gap in a generation* (13) the 62nd World Health Assembly called upon all member states "...to take into account health equity in all national policies that address social determinants of health." (14) Social policy is crucial to breaking the negative spiral linking ill health, poverty and social exclusion. A report commissioned by the European Commission's Employment, Social Affairs and Equal Opportunities Directorate (DG EMPL) establishes the economic and social costs of not having social policies of adequate quality and shows that more equity can go hand in hand with greater efficiency. The report concludes that social policies based on investment in human and social capital are conducive to higher economic efficiency as they improve productivity and quality of the labour force. (15)

Challenges regarding evidence on the cost effectiveness of action

The Wanless Report to the UK Treasury highlighted the need for more economic evaluations of health promoting policies, as the absence of such evidence is considered to be a significant barrier to achieving commitment by governments. (16) However conducting economic evaluations of policies affecting social determinants of health and health inequalities raises a number of methodological challenges because policies and interventions addressing public health issues generate very broad costs and benefits. A review of studies conducted in this area identified four main challenges: attributing outcomes to interventions; measuring and valuing outcomes; incorporating equity considerations; and identifying intersectoral costs and consequences. (17) A briefing paper which reviewed the current state of economic appraisals of public health interventions found that even where outcomes can be attributed to interventions, they usually incur a long time lag which is at odds with most decision-makers preference for benefits that occur quickly. Furthermore, unlike many clinical interventions, public health interventions often require a change in individual or population behaviour in order to be effective, therefore the behaviour change itself must be included in the economic analysis. Given the challenges associated with undertaking economic evaluations of health promoting policies, the authors assert that "...there is presently insufficient economic evaluation evidence to knowledgably inform public health policy making." They suggest however that "...this state of affairs can be changed, but will require strong direction to ensure that priorities for economic evaluation evidence become organised and coordinated at local, regional and national levels." (18)

Calculating the distributional costs and benefits of public health interventions across different sections of the population adds another layer of complexity to the conduct of economic evaluations. In practice many economic evaluations do not consider the impact of the intervention on different population groups focusing instead on the whole population in their analysis. Even where it is known that the impact of a particular intervention may be unequally spread across population groups, it remains challenging to incorporate this into an economic evaluation.

³ The term Health Systems includes all actors, organisations, institutions and resources which have the potential to promote health in society

Conclusion

Economic arguments for cross sectoral investment in health are gaining momentum across Europe and there is increasing appreciation of the economic benefits that may accrue if action is taken to reduce health inequalities. While difficult to undertake, efforts are being made to analyse in economic terms the impact on health and health inequalities of a range of policies outside of healthcare. For example, the Global Health Equity Group (Marmot review) is currently undertaking a strategic review of health inequalities in England post 2010 which incorporates the economic perspective. (19) One of the contributors, the Centre for Health Economics in England, has recently published *An economic framework for analysing the social determinants of health and health inequalities* which includes a review of economic approaches to evaluation and priority setting for policies aimed at reducing health inequalities. (20)

By providing initial evidence, the reports highlighted here can pave the way for further exploration at EU and member state level to analyse in economic terms the health impacts of policies that address the social determinants of health. Such work can help to provide further evidence to support the argument that targeted investment to address health inequalities by action on the social determinants of health is more cost effective than paying for the consequences of health inequalities.

3. Opportunities and challenges to using economic arguments

Introduction

This chapter examines the opportunities and challenges to using economic arguments to address social determinants of health inequalities, as reported by DETERMINE partners. Many influencing factors were described and these have been grouped into six areas:

- Support for addressing social determinants of health inequalities.
- Acceptability of using economic arguments to achieve better health outcomes.
- Clear and meaningful arguments.
- Leadership from health ministries.
- Availability of specialised personnel, data and techniques.
- Consideration of health in assessment procedures.

Support for addressing social determinants of health inequalities

Findings from the first year of the project indicated that relatively few countries have government policy which emphasises social determinants of health inequalities. In the absence of policy support, sectors outside of health may be less likely to routinely or systematically consider health impacts of their policy. In the first year of the project, it was reported that potential health impacts are often not recognised in policies formulated by sectors such as transport, agriculture and housing. (21)

A pattern was observed between the existence of government policy which recognises social determinants of health and health inequalities and the routine inclusion of health outcomes in economic evaluations of non-health policy. In countries where there is strong policy support for a health in all policies perspective, it appears more likely that health outcomes will be considered in economic evaluations of non-health policy. On the other hand, in many countries general awareness about social determinants of health remains low and sectors outside of health do not recognise or place a value on the health outcomes of their policies. Therefore when economic evaluations are conducted on these policies, it appears that potential costs and benefits to health are less likely to be factored into the analysis.

Acceptability of using economic arguments to achieve better health outcomes

The premise that economic arguments should be used to support action on social determinants of health inequalities attracted mixed responses. A number of partners suggested that economic arguments were not routinely used because of a lack of acceptability of such arguments. Resistance to using economic arguments in these cases was based on the following reasons:

 General moral/ ethical arguments, for example "...the focus should be on producing health not saving money" and "...the purpose of the state is not income generation but to ensure better quality lives for its citizens."

- The long lag time between intervention and return in many health promoting interventions may lead to such interventions being considered poor value for money particularly if a short or medium term perspective is taken.
- Interventions with some population groups may require higher initial investment or may produce less substantial economic returns and thus may be considered poor value for money when compared with the general population.
- Economic arguments may be perceived as not relevant or unnecessary, for example where there is a high standard of living or where personal responsibility for health is highly valued.

The legitimacy of using economic arguments is addressed in the Norwegian report *Health Effects in National Economic Analyses*. It is particularly concerned with the equity perspective and recommends that "...applying economic value on lives and health is only relevant in overall economic analyses. This type of assessment should not be applied to specific treatments or interventions. It is important that economic assessments of interventions and treatment are not conducted at the individual level." (22) This is particularly relevant to the equity concern raised by DETERMINE partners regarding distribution of finite resources amongst different groups within the population.

The report also makes reference to the converging views of relevant ministries "…in the past the Ministry of Finance has used economic arguments and the Ministry of Health has used health arguments. Nowadays the debate seems to be more balanced and both arguments are considered relevant by both partners." (22)

Clear and meaningful arguments

Some partners suggested that economic arguments to support a health in all policies approach need to be made clearer and presented in language that speaks to policy makers across a range of sectors. For example, in Scotland the report of the Ministerial task force on health inequalities *Equally Well* states that "...a reduction in health inequalities, by improving the health of those most deprived, is likely to result in a reduction of costs to the National Health Service (NHS) and society as a whole." (23)

Explicitly stating the economic cost of ill health or the savings made to the economy by retention of good population health was also considered to be useful. Such costs have been calculated for several causes of ill health in Scotland including alcohol, smoking, obesity, mental illness, teenage pregnancy and asthma. Of these, mental illness is estimated to incur the highest costs $- \pm 1.5$ billion per annum to NHS Scotland and more than ± 7 billion per annum in wider economic and social costs. Alcohol related illness is estimated to have cost the NHS ± 110 million per annum while the wider economic, human and social costs amounted to some ± 736 million per annum. (23)

An example from Norway explicitly links participation in the workforce and continued existence of the welfare state. A report by the Ministry of Labour and Social Inclusion makes it clear that the latter is dependent on employee contributions which will only continue if the workforce remains in good health. (24) In Slovenia, the focus has been on geographical inequalities in health. Demonstrating a relationship between poor health indicators and weaker social and economic performance at a regional level has prompted a response from both politicians and professionals. (25)

Leadership from health ministries

While overall government policy sets the context for supporting the economic case, it has been suggested by WHO and others that leadership must come from within health ministries. (12) An example is given from the Netherlands where the Ministry of Health has approached a number of national advisory boards including the Onderwijsraad (Education Advisory Board), ROB (Board for Public Government), RVZ (Council for Public Health and Health Care) and the SER (Social Economical Advisory Board), and asked them to identify areas of mutual interest with health. (26) Other indications of leadership within this Ministry include the hosting of a conference in November 2008 entitled "Health is Wealth". (27)

Availability of specialised personnel, data and techniques

Reflecting the findings of the literature review, many partners commented that the complexity of undertaking economic evaluations was a barrier to their conduct. A significant proportion of economic evaluations cited in chapter four were conducted on policies which already have better health as an explicit objective, for example, those related to smoking cessation or diabetes prevention. Fewer examples were reported of economic evaluations which considered health outcomes of policies addressing the social determinants of health such as those related to transport or education. A number of partners suggested that where social determinants of health are not recognised by those undertaking the evaluations, health outcomes are less likely to be considered. Even where the policy has explicit health outcomes, economic evaluations may be more likely to focus on individual determinants of health than social causes. Furthermore they may not always consider indirect or longer term outcomes, thereby reducing the calculated cost effectiveness of an intervention.

It was also suggested that the evidence base on social determinants of health is not as strong as can be found for clinical or medical interventions due to difficulty in determining effects. This is recognised in an example from the Republic of Ireland which notes difficulties in accurately costing health benefits. In the *Economic Evaluation of Water Supply and Waste Water Projects* the authors state that "...there may be some health benefits due to the improved water quality, but these are likely to be difficult to quantify." (28)

Some partners reported that there is disagreement amongst experts on the methodology used to evaluate health outcomes in economic terms. This is highlighted in a Norwegian report *Role of the Health system in Norwegian Society* which states that "...health effects and distributional health effects are complicated to measure and there are limitations regarding what factors are possible to measure. The numbers that are being used and compared are not absolute, but relative. They also serve as indicators and not facts about health, wealth and wellbeing. These are not unambiguous measures and the causal relations are complex. There is also a certain disagreement among researchers regarding what measures are the most appropriate to use." (29)

While acknowledging difficulties in undertaking these types of economic evaluations, a number of partners referred to measures that aim to advance these efforts. For example, in the UK, all public health recommendations produced by the National Institute for Health and Clinical Evidence (NICE) now include an economic appraisal. Additionally guidance has been issued by NICE on how health economic evidence should be collated and analysed. (30) The WHO CHOICE (CHOosing Interventions that are Cost Effective) programme is also contributing to the evidence base by assembling regional databases on the costs, impact on population health and cost effectiveness of key health interventions. (31) In Scotland, a number of alternative outcome measures are currently being explored which attempt to encapsulate the complex factors influencing how cost effectiveness is measured. (32)

Consideration of health in assessment procedures

Some partners reported on efforts to include health in impact assessment and evaluation processes. For example the use of economic assessments as part of impact assessments was identified in *Prescription for a healthier Norway* (33) and further described in *Health Effects in National Economic Analyses* which supports the development of guidelines for economic assessments in the area of health in both health and non-health sectors. Distributional effects are also acknowledged in this report and it is recommended that these are explicitly included in economic analyses; socioeconomic groups are considered a natural unit for this. (22) The *Instructions for Official Studies and Reports*, used by all ministries and their subordinate agencies to clarify financial and administrative consequences of reforms, has recommended that policies are assessed for likely impacts on health. This recommendation has been taken forward and analytical tools such as Health Impact Assessment are being developed which will measure distributional effects of policies focusing on social determinants of health. (34) In Scotland, health impacts are routinely considered in economic evaluations conducted on legislation being taken forward by the Health Directorate, for example the smoke free public places legislation. (35)

A range of other assessment procedures were reported which currently do not appear to systematically consider health but may present opportunities to do so. Regulatory Impact Assessment/Analysis (RIA) is a tool used to assess the likely costs, benefits and impacts of new regulations. In Scotland, all new legislation requires a Regulatory Impact Assessment. Examples of RIAs conducted on a broad range of policies are available. (35) In the Republic of Ireland, the introduction of Value for Money Reviews and Regulatory Impact Analysis into the Irish Public Service and other Public Sector Agencies has resulted in an increase in the number of economic analyses being conducted in recent years. RIA was introduced in Ireland in June 2005 as part of an overall move towards better government regulation. (36)

At EU level, many policies and spending programmes are required to undergo some form of economic analysis through Impact Assessment and/or Evaluation. These aim to measure progress, efficiency and effectiveness of the measures being introduced. Since economic and social cohesion and solidarity are amongst the main objectives of the EU as laid down in Article 2 of the European Community Treaty, all broadly relevant EU policies and programmes should be assessed as to whether they contribute to this goal. However, where impact on social distribution is not an explicit concern of the policy being introduced, its impact on different socioeconomic groups is unlikely to be evaluated.

Impact Assessments (IA) analyse likely impacts of a policy in the economic, environmental and social fields, outline the advantages and disadvantages of each option and examine possible synergies and trade-offs. However, a scan of some recent IAs revealed that the results of economic analysis is often provided in broad terms, and that most do not go into detail about the projected cost/benefits or cost/efficiency in monetary terms.

As a general rule, all major policy initiatives and legislative proposals on the European Commission's Annual Legislative and Work Programme (CLWP) are required to undergo Impact Assessment. Some other proposals, which do not feature in the CLWP but which potentially have significant impacts, may also require an impact assessment. The precise scope of application is decided on an annual basis. (37)

Most European Commission (EC) policies undergo some form of evaluation, which includes economic analysis, however each major EC spending area appears to have its own evaluation guidelines. Furthermore even areas with clear evaluation guidelines may not always follow these. For example, while it is recommended that Cohesion (Regional) Policy assistance should be subject to regular and rigorous evaluation before, during and after implementation, there appears to be a lack of economic assessments in evaluations actually completed. This is confirmed by the *Study on the Use of Cost-effectiveness Analysis (CEA) in EC's Evaluation*, which found that CEA is seldom applied due to a lack of appropriate data to conduct

them. It identified the main prerequisites for applying CEA: possibility to concentrate on a single affect or available approaches to combine several effects; recognised indicator or rating grid for each effect; possibility to quantify/rate effects; possibility of comparing several interventions/options/scenarios. Since one or more of these prerequisites are missing in many instances, it is not surprising that CEA is far from frequent in EC evaluations. (38)

Conclusion

This chapter has considered the opportunities and challenges identified by partners to using economic arguments to address social determinants of health inequalities. While these have been grouped into six areas in reality most of these issues do not exist in isolation from each other. The synergy between a policy supportive environment for health in all policies and the use of economic arguments to support this approach demands particular attention. For example, the greater availability of specialised personnel, data and techniques may encourage more economic analysis and as more economic analysis occurs, techniques and data become more refined and precise.

Findings presented here indicate that this is a complex area and one that is not systematically incorporated into policy development or implementation. It is clear however that there is increasing interest in having a greater priority and focus attached to economic analysis of policy that impacts on the social determinants of health and that some useful work is emerging in this area. It is worth giving consideration as to how work undertaken by WHO and other international bodies could be adopted within member states. Furthermore action in a number of member states should be monitored and further explored to assess the elements of work amenable to adoption in other countries.

4. Examples of economic evaluations that consider health outcomes

Introduction

This chapter explores the extent to which identified economic evaluations consider health outcomes. They have been grouped into two sections. The first section considers evaluations of policies that have an explicit focus on health behaviours and health outcomes, namely general health promotion interventions, physical activity, diet, obesity, smoking, alcohol and mental health/ suicide. The second section considers evaluations of policies which are concerned with other determinants of health, namely education, agriculture, transport and transport safety, home and leisure safety, climate change and energy efficiency. It should be noted that these examples are based solely on the reports submitted by DETERMINE partners which, as outlined in chapter one, was not intended to be a comprehensive review of all policies in all sectors. It is beyond the scope of this task to assess the extent to which examples reported and issues highlighted reflect overall activity in this area in the different regions and countries.

Health behaviours and outcomes

General health promotion interventions

In Finland a report examining the cost effectiveness of different health promotion actions concluded that, on the basis of the available evidence, many actions are cost effective. The report is available in Finnish with an English summary. (39)

In Belgium (Flemish region), several cost effectiveness studies have been carried out, primarily in the areas of diet, alcohol and tobacco consumption and physical activity. A report is available in Flemish only. (40)

At EU level, a cost comparison of different cardiovascular prevention strategies was undertaken to determine how cost effective population based dietary guidelines were compared to other methods across EU member states. Estimated costs per life year gained were:

- £14 £560 for population-based healthy eating guidelines.
- £300 £790 for smoking cessation.
- £900 minimum for nurse screening and lifestyle advice.
- £6200 £11300 for drug therapy (simvastatin). (41)

Physical activity

In Northern Ireland, the Department of Health, Social Sciences and Public Safety undertook a cost benefit analysis of the *Physical Activity Strategy for Northern Ireland*. The author calculated that successful implementation of the strategy's target to reduce the sedentary population from 20% to 15% could save 365 lives per year or 121 lives if the over 75s were excluded. Based on the latter figure, the value of the associated economic benefit would be \pm 131 million and the direct cost saving to the Northern Ireland health services would be \pm 620 thousand annually. (42)

A cost benefit analysis of walking and cycling track networks in Norwegian cities estimated the benefits of such tracks to outweigh costs by a factor of 4.5:1. The report is available in Norwegian with an English abstract. (43) Also in Norway, the Directorate for Health has produced an economic analysis of the effect of increasing physical activity in the population. The report is available in Norwegian only. (44)

In Finland, a report entitled *Money flows in sport in Finland* concluded that recreational sports/health promoting physical activities have positive health effects on the population and at the same time physical inactivity is very costly to society. The report is available in Finnish with an English abstract. (45)

Diet

In Norway, an economic analysis was conducted on a free fruit and vegetable scheme to pupils in primary and secondary schools which included an analysis of distributional effects. At the outset of the scheme, intake of fruit and vegetables showed a social gradient, where children with parents of low socioeconomic status ate less fruit and vegetables than children with parents of higher socioeconomic status. The report is available in Norwegian only. (46)

At EU level an Impact Assessment was conducted on the *Proposal for a Regulation ... in order to set up a School Fruit Scheme.* It cited scientific evidence which suggests an extra portion of fruit and vegetables per day reduces the risk of cardiovascular disease, diabetes and other diet related diseases by 30%. The resulting healthcare savings over time were then compared with the initial investment of fruit and vegetable provision. Initial estimates suggest substantial savings but the report concludes that more systematic monitoring and evaluation is needed to definitively establish cost effectiveness. (47)

Obesity

In the Republic of Ireland, the Report of the Taskforce on Obesity extrapolates estimates from UK data and applies it to Ireland to estimate that the annual economic cost of obesity in Ireland is in the region of \in 2.7 billion. This is based on an estimated 2000 premature deaths annually attributable to obesity. (48)

In the Netherlands, the National Institute for Health and the Environment (RIVM) produced a report which considers the cost effectiveness of exercise and healthy diet counselling among people with increased risk of type II diabetes mellitus due to overweight and obesity. Four packages of lifestyle counselling were modelled and each predicted substantial savings. The report is available in Dutch with an English abstract. (49)

Economic impacts were calculated as part of the Impact Assessment conducted on the European Commission White Paper on a Strategy for Europe on Nutrition, Overweight and Obesity related health issues. It cited a study in 2005 which estimated direct and indirect costs of obesity at 0.3 % of GDP for the EU15. Extrapolating this to the EU25, the cost of obesity rises to \leq 40.5 billion a year. It should be noted that these calculations do not include costs to industry, the environment or society. (50)

Smoking

A health and regulatory impact assessment was conducted on the proposed smoking ban in public places in Northern Ireland which presented the costs and benefits of three different levels of implementation. Economic analysis identified that the primary resource savings to the NHS from this legislation would be from the reduction in hospital costs associated with treating main diseases linked to active and secondhand smoking i.e. lung cancer, ischaemic heart disease and stroke. Substantial cost savings were estimated over a 30 year period, particularly if Option 3, a comprehensive smoking ban, was introduced. (51) In Scotland, a review of the health and economic impacts of the proposed smoking regulation drew similar conclusions. (52)In Finland, the impact of pricing policy and public interventions to reduce smoking have been analysed from an economic perspective with some consideration given to the distributional effects of interventions. (53)

Alcohol

A Factsheet on Alcohol related harm in Europe produced by the European Commission's DG SANCO estimated that alcohol related harm cost the EU economy \in 125 billion in 2003, equivalent to 1.3% of

GDP. This estimate includes, amongst others, losses due to underperformance at work, work absenteeism and premature death. The actual spending on alcohol related problems in the EU is estimated at about \in 66 billion which includes spending related to crime, traffic accidents, health and disease treatment and prevention. (54)

Mental health/ Suicide

A limited economic analysis was conducted on the Suicide Strategy in Northern Ireland. For 2004, the total estimated cost of suicide to Northern Ireland, including direct, indirect and intangible costs, was £202 million or $\pounds 1.4$ million per suicide. Self-harm also has a significant economic impact in Northern Ireland, accounting for 1.5% of all hospital admissions over the previous five years. These incidents have been calculated as costing the economy $\pounds 6.6$ million due to lost earnings, hospital costs and other lost output. (55)

The annual economic costs of depression have been estimated at $\in 118$ billion across the EU. Direct costs, i.e. healthcare costs, account for only a minor part of this burden. The majority of costs (65%) arise indirectly from loss of productivity, i.e. due to sickness absence, early retirement and premature mortality due to suicide. (56) A systematic review of the evidence from economic evaluations on the prevention of mental illness found that robust evidence on cost effectiveness is limited to a very small number of interventions with restricted scope for generalisability and transferability. The most favourable results are related to early childhood development programmes. (57)

Other social determinants of health

Education

International research suggests significant returns on investment in early childhood education with higher returns more likely where programmes focus on children from lower socioeconomic groups. (58) However this finding should be interpreted with careful reference to the local context. For example, in the Netherlands there are large differences in the types of early childhood care and education offered. On the one hand, crèches are mainly provided to give parents the opportunity to work and may not always have a strong educational approach. On the other hand, preschools generally have a strong educational approach but because of their limited hours of opening, there are few employment benefits to parents. Furthermore it must be noted that benefits are highly dependent on the quality of supply and greater benefits usually incur higher costs. (59) Work has also been conducted in the Netherlands on the costs and benefits of early school leaving. In three out of four interventions studied, investment strongly outweighed the cost, particularly where investment was made in pre-school education. (60)

Agriculture

Economic analysis and future scenarios have been conducted in Slovenia to assess the impact of the Common Agriculture Policy (CAP) on Slovene agriculture. (61)

Transport

The voluntary sector in Scotland has used economic arguments, including health economic arguments, to campaign for more sustainable forms of transport. For example, it is estimated that if 40% of all short journeys were switched from car to bicycle, this would result in a saving of at least $\pounds 2$ billion per year due to reduced mortality and closer to $\pounds 4$ billion per year when improved health is included. (62)

Transport safety

In Scotland, the Scottish Transport Appraisal Guidance includes guidance on how to assess the costs of transport related accidents. (63)

An economic evaluation was conducted on the government road safety strategy in the Republic of Ireland. It considered the costs of fatal and non fatal accidents and estimated the savings brought about by the strategy to be over \in I billion for the prevention of fatal accidents and \in 502 million for non fatal accidents. (64) Conversely the Value for Money Review of the Railway Safety Programme 2004-2008 found that in cost benefit terms, the safety dividend was less than the cost of the investment. (65) General guidance on cost benefit parameters for transport project appraisal is also available. (66)

A report prepared by the European Transport Safety Council examined the cost effectiveness of a number of transport safety measures and considered the following most promising: daytime running lights; random breath testing; audible seat belt reminders; use of European New Car Assessment Programme (EuroNCAP) as an incentive for developing safer cars; and road safety engineering. (67)

Home and leisure safety

The Ministry of Social Affairs and Health in Finland estimated the total costs of home and leisure accidents to society as $\notin 2.5 \cdot \notin 4$ billion. Almost half of the direct costs were health care expenses, one fifth were income transfers due to incapacity to work and almost one fifth were expenses of police or rescue services. The costs of prevention, information and research concerning home and leisure accidents were estimated to be less than half of one percent of the direct costs. The report is available in Finnish with an English summary. (68)

Climate change

The EU target for reduction in domestic greenhouse gas emissions is 20% by 2020 (from 1990 levels). To support a stronger EU climate change policy, calculations were made on the economic benefits to health of a 30% reduction compared to 20%. The findings show that while there are significant economic benefits to be realised from achieving the 20% target (between $\in 13$ billion and $\in 52$ billion), raising the target to 30% is estimated to yield a return of between $\in 20$ billion and $\in 76$ billion in the year 2020 alone. These returns would result from cleaner air and therefore fewer respiratory and health related problems, which would prevent 5,300 cases of bronchitis and 2,800 hospitalisations annually. These benefits would accrue year on year. (69)

Energy efficiency

It has been estimated that cold related illnesses cost the health sector ± 30 million every year in Northern Ireland. (70) However health impacts were not explicitly considered in a value for money review of the *Warm Homes Scheme* in Northern Ireland. While it identified a number of elements in the scheme which were not cost effective in terms of energy efficiency, this was not analysed in health terms. (71) At EU level, an Impact Assessment on the *Proposal for a Directive on energy performance of buildings* found that "...the investment requirements and the administrative costs of the measures that were analysed are relatively low compared to the benefits and the returns. The overall benefits for society in terms of reduction of energy consumption and thus reduced CO2 emissions, energy import dependencies, job creation, especially at the regional and local level, positive health and labour productivity far exceed the costs of the measures analysed." (72)

Conclusion

This chapter has presented examples of economic evaluations conducted in a range of different areas outside healthcare. A number of partners reported difficulty in finding any such examples, suggesting that for many countries, this type of analysis is not common practice. Overall there were more reports of economic analysis conducted on policies which focus on direct health and health related behaviour measures (e.g. smoking) than on policies which focus on the determinants of health (e.g. education). Evidence of distributional analysis was limited; only one example (diet) looked at the impact of policy on different groups.

5. Conclusion and recommendations

Conclusion

The purpose of this task was to explore if and how economic arguments are being used to support action at the EU level and in 12 countries and regions across Europe. It was anticipated that findings could be used to achieve more widespread adoption of effective and proven approaches to tackling inequalities.

The literature provides strong economic arguments for investing in health at population level and also suggests that benefits of reducing health inequalities are beginning to be recognised in economic terms. Furthermore, the recent Communication from the European Commission *Solidarity in Health: Reducing health inequalities in the EU* is another indication of a focus on health inequalities at EU level. (73)

Findings reported in chapters three and four demonstrate that economic arguments can and are being used to address the social determinants of health inequalities. The target audience for such arguments has been identified as mainly EU and national level policy makers, to encourage them to invest in health and/or consider the health and health equity impacts of policies. The primary organisations driving such arguments, particularly at the EU level are WHO Europe and DG SANCO, as well as advocacy groups and independent researchers.

Opportunities and challenges have been explored under six themes in chapter three and serve to illustrate that a number of approaches can be used to reach the common goal of healthy public policy and a reduction in health inequalities. Moreover the synergy created when two or more factors are present appears to be particularly effective.

Examples cited in chapter four reflect the range of policy areas which have considered health in economic terms. However it has been difficult to find a substantial amount of evidence at member state level. Moreover the diversity of responses received suggests that very different situations exist in different countries. While a few respondents gave examples which highlighted some interest and experience in this area, most partners reported that there appeared to be a general lack of appreciation of how economic evaluations could be used to strengthen policy in favour of health and health inequalities. Evidence of economic evaluations that calculate distributional effects appear to be even more limited. Issues such as lack of technical expertise, availability of appropriate data, policy support and leadership are all cited as contributing factors.

General recommendations

This work suggests that there is potential for wider dissemination of what is currently known. Moreover further exploration and development is warranted to achieve more consistent and systematic use of

economic arguments to support action on social determinants of health inequalities. Within member states, Health Ministries should play a leadership role in working to achieve greater permeability and acceptance of this approach across systems and this should be supported by Finance Ministries. This should be replicated at the EU and international levels, with agencies responsible for health taking a lead role, supported by relevant assessment, evaluative and financial mechanisms.

Recommendations for EU institutions

Having access to a consistent and systematic information system on health inequalities across the EU would greatly contribute to enabling economic assessments to incorporate health equity into their analysis. EU institutions and agencies should work with EU member states to establish a common data and knowledge base to measure, monitor, evaluate and report on health inequalities, as set out in the EC report *Solidarity in health: Reducing health inequalities in the EU*.

The actions outlined in the EC report Solidarity in health: Reducing health inequalities in the EU particularly those aimed at building commitment across sectors and developing the contribution of EU policies should, where relevant, incorporate a focus on economic analysis, to clarify the economic rationale for addressing the social determinants of health inequalities.

The EU should establish common approaches and methodologies to undertaking economic analysis that incorporate health and health equity. The 7th EU Framework Programme for Research should include economic research analysis in health equity and health systems performance assessment as well as policy intervention assessment.

The EU should use its internal collaborative mechanisms on health, social and economic policies to ensure health equity indicators are systematically included and applied in EC decision making processes. The outcomes of these assessment and evaluation procedures should be made available on EU websites such as Europa as well as through relevant information systems at member state level.

- As well as a greater focus on health and health equity overall, the EU Impact Assessment procedure should consider economic impacts in these terms.
- Cross sectoral bodies such as the Social Protection Committee and those involved in the Open Method of Coordination should promote the need for and apply the outcomes of economic assessments in their considerations.
- At EU level, common evaluation guidelines for all spending areas should include consideration and quantification of health impact.
- The Council of Ministers and the European Parliament should increasingly seek and apply health equity economic assessments in its decision making.
- EU agencies such as the EU Foundation for Living & Working Conditions should be empowered to include health economic analysis against their range of indicators in reports.
- Eurostat should ensure appropriate indicators for economic analysis are included in data gathering.
- Economic evaluations of initiatives undertaken under EU funding programmes should look at effects on different population groups.

Recommendations for member states

A greater proportion of health systems research funding should be allocated to improving the data and methodologies required to develop appropriate systems and carry out such economic analyses.

Greater priority should be placed on incorporating health and differential impacts into economic analyses.

Health Ministries should take a lead role in identifying and promoting economic arguments for health, public health and health inequalities and should support other ministries in ensuring health and health equity is taken into account from an economic perspective.

Finance Ministries should support this by requiring that economic impact assessments be undertaken as part of budget allocation processes.

Economic analyses should include as broad a number of stakeholders as possible, particularly those directly affected, to ensure that all possible effects are taken into account. This includes the private sector, where economic benefits may be insufficiently taken into account by public sector analysis alone.

Action should also be supported at the regional and local level, particularly cooperation between relevant research institutes, health agencies and municipal authorities to build knowledge on economic impacts of innovative local approaches. Community development initiatives could incorporate information on collective economic effects as well as effects on different population groups in their evaluations.

Recommendations for international organisations

Common comparative international data and indicators on health inequalities to which economic analysis can be applied should be developed. Particularly in the WHO Europe Region, there should be cross reference to common indicators being developed across EU member states.

Systematic cooperation within and between international sources of information on economic analyses that incorporate health and health equity (e.g. the Organisation for Economic Cooperation and Development, WHO, Eurostat, the United Nations Economic and Social Council and the EC) should be enhanced and made more transparent and accessible. Easily accessible databases should be established within and between agencies to allow maximum possible knowledge transfers. The role of international non-governmental networks such as the EU Health Policy Forum, the EU Social Platform, as well as Consumer, Environment and Development platforms should be recognised and supported in gathering and disseminating wide ranging stakeholder-based information relating to economic analyses.

References

- 1. Report of the Commission on Macroeconomics and Health. Macroeconomics and Health: Investing in health for economic development. Geneva:World Health Organization; 2001.
- 2. Suhrcke M, McKee M, Sauto-Arce R, Tsolova S, Mortensen J. The contribution of health to the economy in the European Union. Brussels: European Commission; 2005.
- 3. Personal communication. Comments received with thanks from Prof. M. Suhrcke 8/9/09.
- 4. Byrne D. Enabling good health for all: a reflection process for a new EU health strategy. Brussels: European Commission; 2004.
- 5. Commission of the European Communities. Together for Health: A strategic approach for the EU 2008-2013. Brussels: European Commission; 2007.
- 6. Report from the High-Level Group. Facing the challenge: The Lisbon strategy for growth and employment. Luxembourg: European Commission; 2004.
- 7. Suhrcke M, Sauto Arce R, McKee M, Rocco L. The economic costs of ill health in the European Region. Copenhagen: WHO Regional Office for Europe; 2008.
- Suhrcke M, Nugent RA, Stuckler D, Rocco L. Chronic disease: an economic perspective. London: Oxford Health Alliance; 2006.
- **9.** Mackenbach JP, Meerding WJ, Kunst AE. Economic implications of socioeconomic inequalities in health in the European Union. Luxembourg: European Commission; 2007.
- Report of the Eurothine project. Tackling health inequalities in Europe: An integrated approach. Rotterdam: University Medical Center; 2007.
- Report of the Research Working Group. Inequalities in health/The Black report. London: Department of Health and Social Security; 1980.
- 12. World Health Organization. WHO European Ministerial Conference on Health Systems "Health systems, Health and Wealth". Conference proceedings available at: <u>www.euro.who.int/</u> <u>healthsystems2008</u>
- Report of the World Health Organization Commission on Social Determinants of Health. Closing the gap in a generation. Geneva: WHO, 2008.
- 14. 62nd World Health Assembly. Conference proceedings available at: <u>http://apps.who.int/gb/ebwha/</u> pdf_files/A62/A62_R14-en.pdf
- 15. Fouarge D.The costs of non-social policy:Towards an economic framework of quality social policies and the costs of not having them. Brussels: European Commission (DG EMPL); 2003. Report available at: http://socialpolicy.ucc.ie/EU-docs-socpol/Fouarge_costofnonsoc_final_en.pdf
- 16. Wanless D. Securing good health for the whole population. London: HM Treasury; 2004.
- 17. Drummond M. Assessing the challenges of applying standard methods of economic evaluation to public health programmes. Public Health Research Consortium (D-105); 2006.

- Kelly M, McDaid D, Ludbrook A, Powell J. Economic appraisal of public health interventions. London: Health Development Agency; 2005.
- 19. Global Health Equity Group (Marmot Review). Strategic review of health inequalities in England post 2010. University College London. Further information at: www.ucl.ac.uk/gheg/marmotreview/Documents
- 20. Centre for Health Economics Research paper 52. An economic framework for analysing the social determinants of health and health inequalities (2009). University of York.
- 21. DETERMINE Working Document I. Policies and actions addressing the socioeconomic determinants of health inequalities. Examples of activity at EU and member state level in Europe; 2008. Report available at: www.health-inequalities.eu
- 22. Sælensminde K. Helseeffekter i samfunnsøkonomiske analyser. (Health effects in national economic analyses) Oslo: Directorate for Social and Health Affairs; 2007. Report available at: www.helsedirektoratet.no/vp/multimedia/archive/00020/IS-1435_20969a.pdf
- 23. Report of the Ministerial Taskforce on Health Inequalities. Equally Well. Edinburgh, Scottish Government; 2008. Report available at: www.scotland.gov.uk/Publications/2008/06/25104032/0
- 24. Ministry of Labour and Social Inclusion. Work, Welfare and Inclusion. Report to the Storting No. 9; 2006-2007.
- 25. MURA: Investment for Health and Development Programme. Further information at: www.gken.org/Synopses/Cl_10012.pdf
- 26. Ministerie VWS. Naar een weerbare samenleving. Beleidsplan aanpak gezondheidsverschillen op basis van sociaal economische achtergronden. The Hague, Ministerie VWS; 2008. Report available at: www.ggd.nl/ggdnl/uploaddb/downl_object.asp?atoom=50623&VolgNr=698
- 27. Further information at: <u>www.knmg.artsennet.nl</u>
- 28. DKM Economic Consultants. Economic Evaluation of Water Supply and Waste Water Projects. Dublin, DKM Economic Consultants; 2004. Report available at: <u>www.environ.ie/en/Publications/Environment/</u> <u>Water/FileDownLoad, 1543, en.doc</u>
- 29. Directorate for Health. Skapes helse, skapes velferd helsesystemets rolle i det norske samfunnet. (The role of the health system in the Norwegian Society). Oslo, Directorate for Health; 2008. Report available at: www.helsedirektoratet.no/vp/multimedia/archive/00062/Health_creates_welfa_62299a.pdf
- 30. National Institute for Health and Clinical Evidence. Methods for the development of NICE public health guidance (second edition), London, NICE; 2009. Report available at: <u>www.nice.org.uk/media/2FB/53/</u> <u>PHMethodsManual110509.pdf</u>
- 31. WHO-CHOICE Programme. World Health Organization. Further information at: www.who.int/ choice/en/
- **32.** National Performance Framework. Scottish Government. Further information at: www.scotland.gov. www.scotland.gov"//www.scotland.gov"///www.scotland.gov"///www.scotland.gov"///w

- 33. Ministry of Health and Care Services. Prescriptions for a healthier Norway. Report to the Storting No. 6, 2002-2003. Oslo, Ministry of Health and Care Services; 2003. Report available at: www. regjeringen.no/nb/dep/hod/dok/regpubl/stmeld/20022003/Report-No-16-2002-2003-to-the-Storting-.html?id=452203
- 34. Ministry of Modernisation. Instructions for Official Studies and Reports. (Utredningsinstruksen); 2005.
- 35. Scotland. Regulatory Impact Assessment. Further information at: www.scotland.gov.uk
- 36. Ireland. Regulatory Impact Analysis. Further information at: www.betterregulation.ie
- European Commission. Impact Assessment Guidelines. Brussels: European Commission; 2005. Report available at: <u>http://ec.europa.eu/governance/impact/docs/key_docs/sec_2005_0791_en.pdf</u>
- **38.** European Commission. Study on the Use of Cost-effectiveness Analysis in EC's Evaluations. Brussels: European Commission (DG BUDGET); 2006. Report available at: <u>http://ec.europa.eu/budget/library/</u><u>documents/evaluation/studies/cea_finalreport_en.pdf</u>
- Kiiskinen U, Vehko T, Matikainen K, Natunen S, Arpo A. Terveyden edistämisen mahdollisuudet. Helsinki: Ministry of Social Affairs and Health; 2008. Report available at: <u>http://urn.fi/URN:ISBN:978-952-00-2504-5</u>
- 40. De Smedt D, Bakker M, Annemans L. Geïntegreerd actieplan vording en beweging 2008-2015. Kosteneffectiviteitsstudie. Ghent: Ghent University; 2008. Report available at: <u>www.vigez.be/</u> Conference proceedings available at: <u>www.gezondheidsconferentie.be/uploadedFiles/subsite02/</u> <u>actieplan english summary.pdf</u>
- 41. Brunner E, Cohen D, Toon L. Cost effectiveness of cardiovascular disease prevention strategies: a perspective on EU food based dietary guidelines. Public Health Nutrition 4(2B), 711-715; 2001.
- 42. Swales C.A health economics model. The cost benefits of the physical activity strategy for Northern Ireland - a summary of key findings. Belfast: Health Promotion Agency; 2001. Report available at: <u>http://hpani.org/Resources/physicalactivity/paeconomreport.htm</u>
- 43. Sælensminde K Walking- and cycling track networks in Norwegian cities. Cost-benefit analyses including health effects and external costs of road traffic. Oslo: Institute of Transport Economics (TOI); 2002. Report available at: <u>www.toi.no/article17775-29.html</u>
- 44. Sælensminde K Positive helseeffekter av fysisk aktivitet. En konkretisering av veien mot mer fullstendige samfunnsøkonomiske analyser (Positive health effects of physical activity). Oslo: Directorate of Health; 2008. Report available at: <u>www.helsedirektoratet.no/vp/multimedia/archive/00062/_Positive_helseeffek_62779a.pdf</u>
- 45. Ministry of Education. Money flows in sport in Finland. Helsinki, Ministry of Education; 2005. Report available at: <u>www.minedu.fi/OPM/Julkaisut/2008/Liikunnan_rahavirrat_Suomessa_2005?lang=en</u>
- 46. Sælensminde K. Frukt og grønnsaker i skolen Beregning av samfunnsøkonomisk lønnsomhet (Fruit and vegetables in schools. Estimation of national economic profitability). Oslo: Directorate of Health and Social Affairs; 2005. Report available at: <u>www.helsedirektoratet.no/publikasjoner/rapporter/</u> <u>frukt og gr nnsaker i skolen 50393</u>

- 47. European Commission. Impact Assessment. Proposal for a Regulation ... in order to set up a School Fruit Scheme. Brussels: European Commission (DG AGRI); 2008. Report available at: <u>http://ec.europa.eu/governance/impact/cia_2008_en.htm</u>
- **48.** Report of the Taskforce on Obesity. Obesity, the policy challenges. Dublin: Department of Health and Children; 2005. Report available at: www.dohc.ie/publications/pdf/report_taskforce_on_obesity.pdf
- 49. Bemelmans WJE, Wendel-Vos GCW, Bogers RP, Milder IEJ, de Hollander EL, Barte JCM, Tariq L, Jacobsvan der Bruggen MAM. Kosteneffectiviteit beweeg- en dieetadvisering bij mensen met (hoog risico op) diabetes mellitus type 2. Bilthoven: RIVM; 2008. Report available at: www.rivm.nl/bibliotheek/ rapporten/260401005.html
- 50. European Commission.White Paper on A Strategy for Europe on Nutrition, Overweight and Obesity related health issues Summary of the Impact Assessment. Brussels, European Commission; 2007. Report available at: <u>http://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=SEC:2007:0707:FIN:EN:HTML</u>
- 51. Department of Health, Social Services and Public Safety. Smoking (Northern Ireland) Order 2006. Health and Regulatory Impact Assessment. Belfast: DHSSPSNI; 2006. Report available at: www.dhsspsni.gov.uk/ph health regulatory impact assessment smoking ni order 2006.pdf
- 52. Ludbrook A, Bird S, van Tejlingen E. International review of the health and economic impact of the regulation of smoking in public places. Edinburgh: Health Scotland; 2005. Report available at: www.healthscotland.com/uploads/documents/InternationalReviewFullReport.pdf
- 53. Pekurinen, M. Economic aspects of smoking: Is there a case for government intervention in Finland. National Agency for Welfare and Health, Research Reports 16; 1992. ISBN 951-5738-6.
- 54. European Commission. Factsheet on Alcohol-related harm in Europe Key data by the European Communities. Brussels: European Commission (DG SANCO); 2006. Report available at: http://ec.europa.eu/health/ph determinants/life style/alcohol/documents/alcohol factsheet en.pdf
- 55. Department of Health, Social Services and Public Safety. The Northern Ireland suicide prevention strategy and action plan 2006-11. Belfast: DHSSPSNI; 2006. Report available at: www.dhsspsni.gov.uk/ phnisuicidepreventionstrategy action plan-3.pdf
- 56. Sobocki P, Jonsson B, Angst J, Rehnberg C. The Cost of Depression in Europe. Journal of Mental Health Policy Economics. 9:2: 87-98; 2006.
- 57. Zechmeister, I, Killian R, McDaid D. Is it worth investing in mental health promotion and the prevention of mental illness? A systematic review of the evidence from economic evaluations. BMC Public Health 8:20; 2008.
- 58. Cleveland, G. en Krashinski, M. Financing ECEC services in OECD countries; Paris: OECD; 2003.
- 59. Further information at: www.onderwijsachterstanden.nl
- **60.** Veld R. Kosten en Baten van Voortijdig Schoolverlaten. (Costs and Benefits of Early School Leaving). Eindrapport; 2005. Further information at: <u>www.wrr.nl/content.jsp?objectid=3759</u>
- 61. Lock K. Health impact assessment of agriculture and food policies: lessons learnt from the Republic of Slovenia. In: Bulletin of the World Health Organization 2003; 81 (6). Available at: <u>www.who.int/</u> <u>bulletin/volumes/81/6/en/lock.pdf</u>

- 62. Warren J. Towards a healthier economy. Why investing in sustainable transport makes economic sense. Transform Scotland; 2008. Report available at: www.transformscotland.org.uk/towards-a-healthier-economy.aspx
- 63. Donnelley RR. Scottish transport appraisal guidance. Edinburgh: transport Scotland; 2008. Report available at: www.transportscotland.gov.uk/files/documents/reports/j9760/j9760.pdf
- 64. Goodbody Economic Consultants. Economic Evaluation of the Government Strategy for Road Safety, 1998-2002. Dublin: Goodbody Economic Consultants; 2005. Report available at: <u>www.transport.ie/</u> <u>upload/general/7535-0.pdf</u>
- **65.** Department of Transport.Value for money review of the railway safety programme 2004-2008. Dublin: Department of Transport; 2008. Report available at: www.transport.ie/upload/general/11293-0.pdf
- 66. Goodbody Economic Consultants. Cost Benefit Parameters and Application Rules for Transport Project Appraisal. Dublin: Goodbody Economic Consultants; 2005. Report available at <u>www.rsa.ie/</u> <u>publication/publication/upload/Cost Benefit Parameters and Application Rules for Transport Project</u> <u>Appraisal.doc</u>
- 67. European Transport Safety Council, Cost Effective EU Transport Safety Measures. Brussels: European Transport Safety Council; 2003
- 68. Ministry of Social Affairs and Health Home and leisure accidents and their prevention. Helsinki, Ministry of Social Affairs and Health; 2003. Report available at: <u>http://pre20031103.stm.fi/suomi/eho/julkaisut/tapaturma/summary.htm</u>
- 69. Holland MR. The co-benefits to health of a strong EU climate change policy. Prepared on behalf of Climate Action Network Europe, Health and Environment Alliance. WWF; 2008. Report available at: www.climnet.org/Co-benefits%20to%20health%20report%20-september%202008.pdf
- 70. Bell, M. Towards a fuel poverty strategy for Northern Ireland. NICVA. Report available at: <u>www.nicva.</u> <u>org/index.cfm/section/Responses/key/954C0FE4-B0D0-7815-0F3C198069F9D929</u>
- 71. Northern Ireland Audit Office. Warm homes: tackling fuel poverty. Belfast, Northern Ireland Audit Office; 2008. Report available at: www.niauditoffice.gov.uk/pubs/warmhomes/Warm homes final.pdf
- 72. European Commission. Impact Assessment. Proposal for a Directive on energy performance of buildings (recast). Brussels, European Commission (DG TREN); 2008. Report available at: <u>http://ec.europa.eu/governance/impact/ia carried out/cia 2008 en.htm#tren</u>
- 73. Commission of the European Communities. Solidarity in health: Reducing health inequalities in the EU. Brussels, European Commission; 2009. Communication available at: <u>http://ec.europa.eu/health/ph_determinants/socio_economics/documents/com2009_en.pdf</u>

Appendix 1: Glossary of frequently used terms

Social determinants of health (SDH)

Health is determined by many factors outside the reach of the healthcare sector. The term social determinants of health (SDH) is used to highlight that the social conditions in which people live affect their health. As economic and environmental conditions are often closely linked with social conditions these may also be included in a broader interpretation of the term.

Health inequalities

Health inequalities refer to the variation in health experienced by different groups within a population.

Social determinant of health inequalities (SDHI)

A social determinant of health inequalities (SDHI) perspective combines the two concepts in order to emphasise the role of social conditions in people's different rates of health and illness.

Health in all Policies (HiaP)

A Health in all Policies (HiaP) approach promotes the role of sectors outside of health in taking health and wellbeing of citizens into account in their policies. It uses the SDH concept as a rationale for this approach.

Economic evaluation

Economic evaluation compares the costs and consequences of alternative courses of action.

Appendix 2: Task timeline & questionnaire

Timeline

Date	What	How	Where
May 2008	Scoping	Workshop	Consortium Meeting, Ljubljana
Jun-Jul 2008	Gathering information:	Literature review and con- tact with key informants	Desk based
Aug 2008	Discussion and agree- ment of issues	Workshop	Working Meeting, London
Sept 2008 - Feb 2009	Gathering information	Guidance document sent to all partners	Desk based/ Meetings
Mar-Apr 2009	Analysis	Reviewing partner responses and literature	Desk based
May 2009	Presenting draft findings	Agenda item and general discussion	Consortium Meeting, Prague
Jun-Nov 2009	Working paper	Reviewing and finalising paper	Desk based

Questionnaire

Questions one to three seek particular examples of where economic analyses have been conducted. The first question is broad and seeks examples which may not necessarily acknowledge health impacts in the analysis. The second and third questions seek examples which explicitly acknowledge impacts on health (question 2) and health inequalities (question 3). The remaining questions are intended to explore some of the background and arguments for using economic analyses to strengthen the case for action on the social determinants of health and health inequalities.

- 1. Are economic analyses used in relation to policies that address social determinants of health in your country, such as transport policy, agriculture policy, housing policy and so on? Please provide information on as many examples as possible.
- 2. Have you found evidence of attempts to analyse in economic terms the **health impacts** of policies that address social determinants of health? Please provide information on as many examples as possible.
- Have you found evidence of attempts to analyse in economic terms the impact on health inequalities of policies that address social determinants of health? Please provide information on as many examples as possible.
- 4. What economic arguments are used for policies addressing the social determinants of health and health inequalities in your country? Please give examples as to how these arguments are articulated.
- 5. Who or what organisations or sectors are driving such arguments?
- 6. Who or what organisations or sectors are such arguments intended for?
- 7. If there is no or limited debate around economic arguments for such policies, please comment on why you think this may be the case
- 8. What economic arguments might attract and engage policy sectors outside of health to develop joint initiatives on health and/ or health inequalities?